Author's response to reviews

Title: A cluster randomised trial of a telephone-based intervention for parents to increase fruit and vegetable consumption in their 3- to 5-year-old children: study protocol

Authors:

Rebecca J Wyse (rebecca.wyse@hnehealth.nsw.gov.au)
Luke Wolfenden (luke.wolfenden@hnehealth.nsw.gov.au)
Elizabeth Campbell (libby.campbell@hnehealth.nsw.gov.au)
Leah Brennan (leah.brennan@med.monash.edu.au)
Karen J Campbell (karen.campbell@deakin.edu.au)
Amanda Fletcher (amanda.fletcher@hnehealth.nsw.gov.au)
Jenny Bowman (jenny.bowman@newcastle.edu.au)
Todd R Heard (todd.heard@hnehealth.nsw.gov.au)
John Wiggers (john.wiggers@hnehealth.nsw.gov.au)

Version: 6 Date: 27 April 2010

Author's response to reviews: see over
Editor-in-Chief
Melissa Norton, MD

26th April 2010

Dear Ms Norton,

RE: Formatting changes to study protocol submitted to BMC Public Health

Please find attached the final version of the manuscript ‘A cluster randomised trial of a telephone-based intervention for parents to increase fruit and vegetable consumption in their 3- to 5- year-old children: Study Protocol’.

The manuscript has been proof read for typographical errors and we are happy with the content. We hope it can now proceed to publication.

Kind regards,

Rebecca Wyse